Author’s response to reviews

Title: Peripheral ganglia in healthy rats as target structures for the evaluation of PSMA imaging agents

Authors:

Heike Endepols (heike.endepols@uk-koeln.de)
Agnieszka Morgenroth (amorgenroth@ukaachen.de)
Boris Zlatopolskiy (boris.zlatopolskiy@uk-koeln.de)
Philipp Krapf (p.krapf@fz-juelich.de)
Johannes Zischler (j.zischler@fz-juelich.de)
Raphael Richarz (raphael.richarz@adacap.com)
Sergio Vásquez (sergio.munoz_vazquez@yahoo.com.mx)
Bernd Neumaier (b.neumaier@fz-juelich.de)
Felix Mottaghy (fmottaghy@ukaachen.de)

Version: 2 Date: 10 Jun 2019

Author’s response to reviews:

Dear editorial team,

we have uploaded a clean version of the manuscript (without track changes).

The sentence "The funding body had no role in the design of the study or collection, analysis, and interpretation of data or in writing the manuscript." has been added to the funding section.

The author affiliations have been adjusted according to the new publication guidelines of the University of Cologne and the Forschungszentrum Jülich.

"Ethics approval" has been renamed as "Ethics approval and consent to participate".

Sincerely,

Bernd Neumaier and Heike Endepols